Author's response to reviews

Title: Airway, but not serum or urinary, levels of YKL-40 reflect inflammation in early cystic fibrosis lung disease.

Authors:

Emmanuelle Fantino (e.fantino@uq.edu.au)
Catherine Gangell (c.gangell@uq.edu.au)
Dominik Hartl (dominik.hartl@med.uni-tuebingen.de)
Peter D Sly (p.sly@uq.edu.au)

Version: 2 Date: 9 October 2013

Author's response to reviews:

Tonilynn Manibo
Journal Editorial Office
BioMed Central

Dear Tonilynn,

As requested, the manuscript now states that Informed consent was obtained from the parents of participants at the time of BAL for collection and use of participant samples.

Regards,
Peter Sly